6

https://doi.org/10.1038/s43856-024-00503-0

Household-level effects of seasonal malaria chemoprevention in the Gambia

Check for updates

Seyi Soremekun $\mathbb{O}^1 \boxtimes$, Bakary Conteh², Abdoullah Nyassi², Harouna M. Soumare², Blessed Etoketim², Mamadou Ousmane Ndiath², John Bradley \mathbb{O}^3 , Umberto D'Alessandro², Teun Bousema \mathbb{O}^4 , Annette Erhart², Marta Moreno¹ & Chris Drakeley $\mathbb{O}^1 \boxtimes$

Abstract

Background In 2022 the WHO recommended the discretionary expansion of the eligible age range for seasonal malaria chemoprevention (SMC) to children older than 4 years. Older children are at lower risk of clinical disease and severe malaria so there has been uncertainty about the cost-benefit for national control programmes. However, emerging evidence from laboratory studies suggests protecting school-age children reduces the infectious reservoir for malaria and may significantly impact on transmission. This study aimed to assess whether these effects were detectable in the context of a routinely delivered SMC programme.

Methods In 2021 the Gambia extended the maximum eligible age for SMC from 4 to 9 years. We conducted a prospective population cohort study over the 2021 malaria transmission season covering 2210 inhabitants of 10 communities in the Upper River Region, and used a household-level mixed modelling approach to quantify impacts of SMC on malaria transmission.

Results We demonstrate that the hazard of clinical malaria in older participants aged 10+ years ineligible for SMC decreases by 20% for each additional SMC round per child 0–9 years in the same household. Older inhabitants also benefit from reduced risk of asymptomatic infections in high SMC coverage households. Spatial autoregression tests show impacts are highly localised, with no detectable spillover from nearby households. **Conclusions** Evidence for the transmission-reducing effects of extended-age SMC from routine programmes implemented at scale has been previously limited. Here we demonstrate benefits to the entire household, indicating such programmes may be more cost-effective than previously estimated.

Seasonal malaria chemoprevention (SMC) with sulphadoxinepyrimethamine and amodiaquine (SP-AQ) is a World Health Organisation-endorsed strategy to prevent severe malaria and malaria deaths in young children in living in the African Sahel/sub-Sahel, where most transmission occurs within a few months of the year¹. Since this recommendation, SMC has been implemented in more than 10 countries covering over 20 million children. Trials have shown that SMC reduces clinical malaria in children aged 0–4 years by more than 80% in the first 4 weeks following a dose, and 67% 4–6 weeks following a dose^{2,3}. In updated 2022 malaria control guidelines the WHO recommended control programmes could increase the SMC-eligible age to children above 4 years⁴ to potentially mitigate a potential shift of the burden of severe malaria to older children as malaria transmission declines in endemic settings⁵⁻⁷. This was based on evidence from a 2010 stepped-wedge trial in Senegal showing a reduction of 61% in clinical cases in children 5–9 years receiving SMC^{4,6}. However, more recent studies show mixed impact in this age group^{8,9} and concerns about programme sustainability and maintenance of effective coverage in the current 0–4 age group have led

Plain language summary

Seasonal malaria chemoprevention (SMC) is the provision of monthly, preventative, antimalaria medication to young children at times when they are most at risk of severe disease. Recently the World Health Organisation recommended expanding SMC to children older than 4 years. Older children with malaria typically remain symptomless so the advantages were unclear. However, laboratory evidence suggests this group continues to transmit malaria to others. We conducted a population study in 2021 in 10 communities in the Gambia where SMC was extended to all children up to 9 years of age for the first time. We found household members aged over 9 years were less likely to get clinical disease when most young children in the same household did receive SMC. This suggests an added protection of SMC for those who do not receive it, potentially increasing costeffectiveness.

to reservations about the feasibility and resources required to implement an age increase¹⁰.

Children aged over 5 years however have an increasing risk of asymptomatic infection and elevated gametocytaemia¹¹; recent studies of controlled mosquito feeding experiments¹²⁻¹⁵ and simulation analyses¹⁶ have demonstrated asymptomatic infections in children in this age group are also often the largest contributors to malaria transmission. This suggests extended-age SMC, besides reducing clinical illness in direct recipients, may have wider impacts on transmission to non-recipients^{6,17,18}. For policymakers with finite resources, interventions that maximise impact on both disease transmission and clinical disease burden are prioritised, particularly when the goal is elimination¹⁹. The transmission-reducing impact of vector control interventions are well-described²⁰⁻²² and incorporated into models of cost-effectiveness by design as programme impact is frequently measured in the entire population targeted^{20,23,24}. Vector control programmes are predicted to be highly cost-effective and usually prioritised for implementation^{19,24}. The impacts of SMC programmes are normally measured only in eligible children, thus the true value of extended-age implementation is likely to be persistently underestimated, highlighting a crucial evidence gap for implementation design and policy.

In 2021, the National Malaria Control Programme (NMCP) of the Gambia Ministry of Health recommended increasing the eligible age for SMC to 9 years²⁵. Given the highly clustered nature of malaria transmission, particularly in this context^{26,27}, herd impacts of the SMC programme should be primarily detectable within mixed-age households using household-level statistical approaches. Therefore as part of a larger study of malaria infection and transmission dynamics conducted in 2019-2022, 'The INDIE Trial'28, we conducted a prospective cohort study collecting population data on SMC coverage and malaria burden across communities in Gambia's Upper River Region to assess the individual and household-level impacts of the extended programme. We find that in households where most children had received at least one round of SMC, the incidence of clinical malaria and prevalence of asymptomatic infections is lower in adolescents and adults who are themselves non-eligible for SMC compared to households with lower coverage of SMC in children-these results are robust to adjustment. This suggests transmission effects are detectable in routine programmes and extended-age SMC may have additional cost-benefits beyond reducing disease in target children.

Methods Study site

The study was implemented in ten communities (Njayel, Banni Kunda, Temanto, Bolibana, Fula Mori Bochi, Madina Samba Sowe, Njum Bakary, Sare Demba Dardo, Sare Biram and Tabajang) in the Upper River Region (URR) of the Gambia (Fig. 1). The region has a population of over 250,000 inhabitants and has a distinctly seasonal pattern of malaria transmission, where the most cases occur between June and December, peaking directly following the annual rains between June and September²⁹. The main vector species is Anopheles gambiae sensu latu^{30,31}. During the 2021 season, the mean nightly Anopheles gambiae s.l. and Anopheles funestus catch rate per house in the study communities using Centres for Disease Control light traps rate was 0.97 (standard deviation 1.41) (Supplementary File STable 1). The peak temperature during the transmission season was 35.2 °C in 2021 (Supplementary File STable 2). A region-wide distribution drive of long lasting insecticide treated nets occurred in 2019. For the first time in July 2021, The NMCP extended the upper eligible age for SMC from 4 to 9 years. Eligible children aged 0-9 years could receive up to four rounds of monthly SMC with sulphadoxine-pyrimethamine (SP) and amodiaquine (AQ) administered between August and November 2021 by community outreach teams.

Study design

This prospective observational cohort study was nested within a larger trial measuring the effectiveness of control interventions on malaria infection and transmission, the INDIE Trial (*P. falciparum* Infection Dynamics and

Transmission to Inform Elimination, clinicaltrials.gov reference NCT04053907) and used data collected between 26th July 2021 and 12th January 2022. Three communities (Njum Bakary, Sare Demba Dardo and Sare Biram) were randomly assigned to receive 3 monthly rounds of mass drug administration with dihydroartemisinin-piperaquine between April and June 2021. In two communities (Madina Samba Sowe and Tabajang) inhabitants were screened weekly for fever—'fever screen and treat (FSAT)', those with temperatures above 37.5 °C or a history of fever in the last week received a rapid diagnostic test (RDT) for malaria and positive cases were treated with artemether-lumefantrine. Three communities (Banni Kunda, Temanto and Njayel) were assigned to monthly screening of all residents with ultrasensitive-RDTs, and positive cases treated with artemetherlumefantrine-'mass screen and treat' (MSAT). The remaining two communities (Bolibana and Fula Mori Botchi) were assigned to the control group and received standard control interventions. A programme of community case management of malaria was concurrently initiated in all communities. From 26 July 2021, a clinician and support worker (nurse and community health worker) based in each community managed all suspected malaria cases (passive case detection-PCD).

Study procedures

A series of community-based sensitisation activities took place during the INDIE baseline year (2019) to inform community members about the study and its aims, and all inhabitants were invited to take part. Residents who provided informed consent were enroled during a baseline dry season survey in April-May 2021. Cross-sectional surveys covering all enroled community members were carried out on a rolling basis every 8 weeks during the transmission season. Blood samples collected from the enroled population during all surveys were analysed at the MRC Unit The Gambia (MRCG) in Basse by quantitative polymerase chain reaction (qPCR var-ATS) to detect malaria infections. Study questionnaires were administered during the surveys to all participants and collected address details and GPS coordinates, demographics including age, gender and ethnicity, symptoms of illness, care seeking and any treatments received for each participant and use of insecticide-treated bed nets the night prior to the survey. Identical data were collected from malaria patients identified by PCD visits although diagnosis was done by RDT only. The dates and number of all SMC rounds received for each child under 10 years were collected from SMC cards or by caregiver report. SMC data were collected in a survey after the last (November) round of SMC, and at two additional capture points: during a final cross-sectional survey in January 2022 and following a review of the SMC database by SS and AN, in a dedicated SMC mop-up survey to address data queries and collect data on additional study children not met in previous surveys (10th-15th March 2022). Entomological surveillance was conducted during the 2021 study period in 6 randomly selected households per community. In each household, Centre for Disease Control (CDC) light traps were hung in a sleeping room for three consecutive nights per month; caught mosquitoes were examined to determine species, sex and gonotrophic status (fed status and gravidity).

Study outcomes

All data from PCD visits, cross-sectional and entomological surveys were collected onto secure handheld devices and stored, cleaned and accessed via a password protected REDcap[®] server (Vanderbilt University, Nashville Tennessee) hosted at the MRC LSHTM Research Unit in Fajara. Incidence of clinical malaria was defined as the number of passively detected cases per 100 person-months. For each clinical case, 2 weeks of follow up were removed from the denominator. Malaria prevalence was defined as the percentage of participants sampled who were qtPCR-positive during a late season survey (27th September to the 27th November 2021). qtPCR-positive participants sampled during this survey were categorised into high- and low-density infections by age group. High-density infections were those with density above the median of the natural log asexual parasite concentrations per μ L blood for the age groups (0–4, 5–9, 10–15 and 16+ years).



Fig. 1 | Map of study site. The locations of the ten study communities are denoted by black markers in the Upper River Region of The Gambia. The inset shows the location of the region within the country.

Statistical analysis

The impact of SMC status on the incidence of clinical malaria in eligible children was assessed by fitting shared frailty cox models with gammadistributed random effects specified for person to account for the possibility of repeated clinical episodes within the same person (failures). Where frailty models did not stabilise, robust errors were used instead. To account for ongoing malaria control interventions in each community and background community level risk of malaria, the models pre-specified fixed effects for village ID. Models also included fixed effects for the percentage of all visits to the participant at which an insecticide-treated bed net was used the night before and household ID.

To assess the association between household level coverage of SMC in eligible children and malaria burden in adolescents/adults 10+ years of age, we first defined household SMC coverage in two ways: (i) the percentage of children 0–9 years in each household who received at least 1 round of SMC and (ii) the mean number of rounds of SMC per child. Both definitions were implemented as continuous variables and classed into 'low', 'moderate' and 'high' groups thus: <25%: 25–79%, 80%+ (definition 1) and <0.3, 0.3–2 and >2 rounds/child (definition 2), respectively. The high group cut-offs (80% or >2 rounds) were fixed to be consistent with the optimum minimum level of coverage of child health programmes^{32–34}. The low/medium cut-off values were varied for definition 1 in the range 0–55% in 5% increments and for definition 2 in the range 0–0.9 in 0.1 unit increments and implemented in household level models described in Supplementary File 1 STable 3. The models with low groups of <25% and 0.3 had the lowest Akaike and Bayesian Information

Criteria values (AIC and BIC) and were thus selected as the final low-group cut-offs in all household level models.

The impact of household-level coverage of SMC as defined above on clinical malaria episodes in participants 10+ years of age was then assessed by fitting shared frailty Cox models to account for repeated clinical episodes as previously. The impact of household-level coverage of SMC on malaria prevalence in participants 10+ years of age was estimated using logistic regression models with a household-level random effect. We visualised parasite concentration distributions in participants aged 10+ years stratified by household SMC coverage by the generation of kernel density estimate plots and estimated the impacts of household SMC coverage on the prevalence of high-density infections in this group using logistic regression models with a household-level random effect. Equivalent data in children 0-4 and 5-9 years were limited by fewer malaria infections and smaller cell sizes, therefore the additional household-level impacts of SMC on eligible children themselves were assessed in all eligible children aged 0-9 years combined (STables 7, 8 Supplementary File). Where multilevel models failed to converge, we calculated household-level estimates of malaria burden in this age group and regressed these against household-level SMC coverage using a twostage technique to adjust for individual-level covariates^{35,36}. All individual and household-level models of impact of SMC were implemented in Stata version 17.0 (StataCorp, Texas USA).

All household-level models pre-specified fixed effects for village ID, the percentage of all visits to households at which inhabitants used an insecticide-treated bed net the night before, the household level prevalence



of malaria at a baseline dry season survey in April-May 2021, age in years, the household ratio of children aged 0–9 years to participants aged 10+ years, the total number of household inhabitants and for prevalence models, the week of survey visit.

To assess the effect of spatial clustering of household-level impacts of SMC, an inverse weighted distance matrix was created describing the Euclidean distances (km) between all households in the study site. We calculated the mean incidence rates of clinical malaria per participant aged 10 years and older per enumerated household, and the mean incidence rate in children 0-9 years per household and attached additional features of SMC coverage, insecticide-treated bed net usage, compound size and age composition, and geolocation (latitude and longitude) to each household. Identical datasets were constructed with the overall prevalence of asymptomatic infection per household during the late-season survey for the same age groups. Visual examples of the resulting datasets are shown in Supplementary Fig. 2 (Supplementary File). The distance matrix was used to calculate global Moran indices and z-scores to test for spatial autoregression under the hypothesis that prior regression estimates of the association between household-averaged outcomes for incidence and prevalence of malaria and household SMC coverage were not influenced by estimates in nearby households. Models were adjusted for household level covariates as previously stated - percentage of all visits to households at which inhabitants used an insecticide-treated bed net the night before, the household level prevalence of malaria at a baseline dry season survey in April-May 2021, the household ratio of children aged 0-9 years to participants aged 10+ years and the total number of household inhabitants. Visualisation of spatial data, construction of distance matrices and spatial analysis was conducted using Seaborn, Geoplot, Geopandas and the Esda_Moran libraries in Python (Python Software Foundation, Python Language Reference, version 3.9).

Ethical approval

This study did not collect any new data or implement additional activities not previously covered by approvals already in place for the INDIE study. Ethical approval for the INDIE study was provided by the Government of The Gambia/The MRC Gambia Joint Ethics Committee, The Gambia, and the Ethics Committee of the LSHTM, UK (reference: 16642).

Reporting summary

Further information on research design is available in the Nature Portfolio Reporting Summary linked to this article.

Results

Characteristics of the study sample

Census data for the site (year 2021) was provided by the Health and Demographic Surveillance System (DSS) team for the Basse District Area. The estimated combined population for the ten study communities was 2839 residents in 144 households, of which 2435 participants in 133 households were enroled into the main INDIE study. The present study included 2210 participants (825 aged 0-9 years and 1,385 participants aged 10+ years) in 129 households who were met during the 2021 malaria transmission season and had complete address data (Fig. 2). Supplementary data 1 summarises the characteristics of study households by the categories of coverage of children with 1+ round SMC in children aged 0-9 years, and Supplementary data 2 summarises the same information by categories of the mean number of SMC rounds per child. Most households had at least 1 malaria case over the season and most had inhabitants in each age category, though a smaller proportion of households within the lowest household SMC coverage categories had inhabitants aged 0-4 years (69.2%) compared to the overall proportion (89.9%). These households also tended to be smaller than the average (10 versus 17 mean inhabitants respectively).

Impact of SMC status on malaria in eligible children

Dates of SMC rounds were extracted directly from SMC cards for 68.3% of children, and based on caregiver reports for 31.6% of children who had received at least 1 round of SMC. There was no SMC cards available for children who did not receive SMC. Overall, 76.0% of children aged 0–9 years received at least one round of SMC in 2021, and 20.9% received four rounds. Coverage was highest in children aged 0–4 years (90.6% and 31.5% received SMC at least one and all four rounds, respectively) than those aged 5–9 years (64.0% at least one round, 12.1% all four rounds, respectively) (Table 1).

The incidence rates (IR) of clinical malaria over the 2021 season in children eligible for SMC are shown in Table 2 and Kaplan-Meier plots for clinical malaria by SMC coverage are displayed in Supplementary Fig. 1 (Supplementary File). IRs for clinical malaria in children who received at least 1 round of SMC was 1.58 episodes per 100 personmonths of follow-up, compared to an IR of 2.89/100 person-months in children who received none. This was highly significant in both unadjusted and adjusted Cox models (adjusted hazard ratio 0.44 95% CI 0.21, 0.93; *p* value 0.031). The largest reductions were seen in children who received 3–4 rounds of SMC (adjusted hazard ratio 0.38 95% CI 0.16,

Table 1 Coverage of Season Gambia	al Malaria chemopre	vention in 825 childr	en aged 0–9 years d	uring the 2021 mala	ıria transmission sea	ason (MTS), Upper R	iver Region The
SMC round	% children received SI	MC at each round		% children by minim	Im number of rounds of SI	MC received	
	0-9 years (N = 825)	0-4 years (N = 372)	5-9 years (N = 453)	Minimum rounds	0-9 years (N = 825)	0-4 years (N = 372)	5-9 years (N = 453)
No SMC received	24.0% (198)	9.4% (35)	36.0 (163)				
Round 1 (8–19 August 2021) (<i>n</i>)	52.5% (433)	73.1% (272)	35.5% (161)	At least 1	76.0% (627)	90.6% (337)	64.0% (290)
Round 2 (8–17 September 2021) (n)	59.9% (486)	72.3% (269)	47.9% (217)	At least 2	60.1% (496)	77.2% (287)	46.1% (209)
Round 3 (12–16 October 2021) (<i>n</i>)	50.4% (416)	59.1% (220)	43.3% (196)	At least 3	47.6% (393)	62.6% (233)	35.3% (160)
Round 4 (8–12 November 2021) (<i>n</i>)	42.8% (353)	57.3% (213)	30.9% (140)	4	20.9% (172)	31.5% (117)	12.1% (55)

0.92; *p* value 0.031). Reductions were also significant when stratified by age; hazard of clinical infection was very low in children 0-4 years of age who received 1 or more rounds of SMC in the adjusted model. However, associations between SMC and clinical malaria in children aged 5–9 years although consistent with an impact, did not remain significant after adjustment.

STable 4 (Supplementary File) shows malaria prevalence in eligible children by SMC status. Prevalences were low, and lowest in children aged 0–9 years who received 3 or 4 rounds of SMC (2.6%), and higher in those who received only 1–2 rounds (6.5%). Prevalence in children who received no SMC was 4.4%. No impact of SMC coverage on malaria prevalence remained significant in the adjusted models.

Household level effects of SMC coverage

In Cox models with SMC coverage specified as a continuous exposure, there was an inverse and significant association between clinical malaria in participants 10 years or older and coverage of one or more rounds of SMC in younger children in the same household (HR: 0.99; 95% CI: 0.99, 1.00; p = 0.039) and a close to significant inverse association between mean rounds of SMC per child per household and clinical malaria in older participants (HR 0.80; 95% CI: 0.63, 1.02; p = 0.069). Predicted trends from these models indicate a 10% decrease in hazard for a 10% rise in the percentage of children who received any SMC, or a 20% decrease in hazard for each additional round of SMC received per eligible child.

In categorical models, we similarly observed reduced incidence of clinical malaria in participants aged 10 years and older with increasing household coverage of SMC (Fig. 3). After adjustment, the reductions were significant or close to significant and of similar magnitude when SMC coverage in the same household was moderate (25–79% of children received 1+ round SMC—adjusted HR: 0.60; 95% CI: 0.34, 1.07; p = 0.084), or high (80% or more received 1+ rounds SMC—adjusted HR: 0.49; 95% CI: 0.28, 0.86; p = 0.014) compared to households where coverage was low (<25%) (Table 3). Similar significant reductions in clinical malaria were observed in households defined by mean number of SMC rounds per child (Fig. 3 and Table 3).

There was a consistent trend of reduced clinical malaria in children 0–9 years with increasing household coverage of SMC, however, this trend was not significant in the fully adjusted models - except for children who received no SMC residing in households with otherwise high coverage (STable 6 Supplementary File).

Unadjusted continuous mixed effects logistic models estimated significant reductions in malaria prevalence with increasing household SMC coverage in 1056 participants aged 10+ years visited in the late malaria season (unadjusted OR: 0.98; 95% CI: 0.97, 0.10; *p* = 0.022 and OR: 0.64; 95% CI 0.42, 0.97; p = 0.034 for SMC coverage defined as percentage of children in household with 1+ SMC rounds or mean SMC rounds per child in household, respectively). These effects did not remain significant after adjustment (adjusted OR: 0.99; 95% CI: 0.98, 1.00; p = 0.200 and OR: 0.82: 95% CI: 0.55, 1.23; p = 0.341 for SMC as previously). When SMC coverage was categorised as low, medium or high, malaria prevalence in older participants ranged from 6.8% in high coverage households to 18.8% in low coverage households (Fig. 4a, c). In categorical mixed effects models, malaria prevalence in participants aged 10+ years was lowest in households in the two highest coverage groups (moderate and high groups) and both unadjusted and adjusted models suggested a strong decreasing trend in prevalence with increasing SMC coverage (Table 4). The inverse was observed on parasite densities in infected individuals: as household SMC coverage increased, the distribution of parasite densities skews higher (Fig. 4b, d) and the risk of high-density infections increases four-fold or more in households in higher coverage groups (Table 4).

Malaria prevalence was relatively low in 684 children 0-9 years visited in the same survey (all <6.5% STable 7 Supplementary File). We detected no significant associations between household SMC coverage and malaria prevalence in children 0-9 years in adjusted models.

Spatial impacts of household level SMC coverage

We successfully enumerated 90% (129/144) of all households in the study site. Following regression of mean household IRs or percent prevalence against household level SMC coverage, we assessed the degree of spatial autocorrelation of model residuals using tests of Moran's I. *P* values for all models exceeded 0.05 (0.90–0.99) indicating no spatial clustering of our household level modelling outcomes (Supplementary Fig. 3 Supplementary File).

Discussion

Empirical evidence for the impact of SMC on malaria transmission is limited but has critical implications for programme scale-up¹⁸. In this study, we have applied a household-level modelling approach to assess the effects of extended-age SMC on older household inhabitants in a pre-elimination setting in the Gambia. In adjusted models, we observed a significant reduction in the incidence of clinical malaria in eligible children who received SMC compared to those who did not. We observed that older participants ineligible for SMC in households with higher coverage of SMC were significantly less likely to contract clinical malaria compared to households where SMC coverage was low. In a survey of the population late in the malaria transmission season we also observed fewer asymptomatic infections in older participants in households with higher SMC coverage. There was no evidence of spatial clustering of these impacts.

Whilst the incidence of clinical malaria in SMC-eligible children aged 0–9 years decreased with increasing household level SMC coverage and irrespective of their individual level SMC status, we could not reliably demonstrate this constituted an additional benefit of SMC in our adjusted models (STable 6 Supplementary File) neither did we detect an impact of household level coverage of SMC on asymptomatic infections in SMC-eligible children (STable 7 Supplementary File).

The impacts observed in this study broadly align with the results of a 2008-2010 pilot stepped wedge trial which experimentally increased the eligible age range for SMC in Senegal to 10 years⁶-to our knowledge this is the only other source of data on clinical impact following SMC in this age group. The authors reported a decrease in the incidence of clinical malaria in eligible children (69%) and in older age groups (26%). The study did not consider the impacts of lower coverage or impact on asymptomatic infections, and was implemented with excellent research fidelity and high SMC coverage over the three years of implementation, where 84-93% of children received 3 rounds of SMC. A key question for decision-making is whether similar impacts on malaria transmission are achievable in routinely delivered large-scale programmes. Following an era of efficacy and effectiveness trials of SMC initially demonstrating high programme coverage and large effects in study populations, emerging data from routine SMC programmes in West Africa suggests mixed, or reduced impact^{8,9,32,37,38}. In our study, coverage of children receiving 3 or more rounds of SMC was 49% overall and only 36% of children 5-9 years; however this was the first season in which the NMCP of the Gambia extended the eligible age for SMC, and the coverage may reflect the initial complexities in operationalising the change. It nonetheless suggests transmission impacts are detectable in the context of programmes with moderate implementation strength.

There is an extensive theoretical basis from both laboratory and field studies which suggest the infectious reservoir for falciparum malaria in sub-Saharan Africa is normally maintained by children up to 15 years of age^{12,15,16}, and the expected extended benefits of preventative interventions in this age group should be observable in non-eligible adolescents and adults¹⁸. We justify the use of a household-level models to assess impact of SMC on transmission based on previous research in The Gambia²⁶ and similar low to moderate endemic settings²⁷ which indicate malaria transmission is highly localised within household/family groups³⁹ with negligible contribution from nearby households. This is supported by the null findings from our global Moran tests for spatial autocorrelation in household-level SMC models in the study site. Based on this proposed framework, we hypothesised the mechanism through which herd impacts of SMC arise will be by first reducing infections in the target age group in the same household. We

able 2 | Incidence and hazard ratios for clinical malaria in 825 children aged 0-9 years by number of rounds of SMC received in the 2021 malaria transmission season, Upper River Region, The Gambia

SMC status	Incidence rates (cases/PM)	per 100 person-m	onths	Unadjusted HR (95% CI)	đ		Fully adjusted HR (95%	é CI) p	
	0-9 years	0-4 years	5-9 years	0-9 years	0-4 years	5-9 years	0-9 years	0-4 years	5-9 years
No SMC	2.89 (31/1073.5)	1.57 (3/191)	3.17 (28/882.5)	1 (ref)	1 (ref)	1 (ref)	1 (ref)	1 (ref)	1 (ref)
1+ rounds SMC	1.58 (54/3422)	1.14 (21/1843)	2.09 (33/1578.5)	0.43 (0.27, 0.67) < 0.001	0.45 (0.13, 1.52) 0.196	0.55 (0.33, 0.91) 0.022	0.44 (0.21, 0.93) 0.031	0.08 (0.01, 0.72) 0.024	0.41 (0.13, 1.39) 0.156
1-2 rounds	1.88 (24/1274.5)	1.76 (10/567)	1.98 (14/708)	0.56 (0.32, 0.97) 0.037	0.84 (0.22, 3.16) 0.791	0.53 (0.28, 1.02) 0.056	0.50 (0.23, 1.10) 0.084	0.13 (0.01, 1.44) 0.097	0.29 (0.08, 1.02) 0.053
3-4 rounds	1.40 (30/2147)	0.86 (11/1276)	2.18 (19/870.5)	0.36 (0.22, 0.60) < 0.001	0.31 (0.09, 1.12) 0.074	0.56 (0.31, 1.02) 0.060	0.38 (0.16 0.92) 0.031	0.05 (0.01, 0.43) 0.006	0.60 (0.15, 2.32) 0.460
Showing incidence rat fixed effects.	tes as episodes of clinic	sal malaria per 100 pe	rson-months (PM) of fc	ollow-up. Total cases and total pe	erson-months per group in bra	ckets. Adjusted Cox regressio	n models included % LLITN u	sage over transmission season	, village and household ID as



time (t=days) since start of PCD

Fig. 3 | Incidence of clinical malaria by household-level coverage of SMC: Kaplan-Meier survival plots of clinical malaria episodes in 1385 study participants aged 10+ years by SMC coverage in eligible children aged 0–9 years in the same household. Vertical axes start at '60% no clinical malaria at time = t' for clarity. SMC coverage is defined as (a) percentage of eligible children in household with 1+ rounds of SMC, where the blue line is participants in households with <25% coverage, the orange line refers to those in households with 25–<80% coverage and the green line those with 80%+ coverage. **b** Mean number of SMC rounds per eligible child, were blue line is participants in households with <0.3 rounds/child, the orange line refers to those in households with 0.3–<2 rounds per child, and the green line those with 2+ rounds per child. Grey vertical lines indicate the first reported day of each monthly round of SMC.

Table 3 | Incidence rates and hazard ratios (HR) for clinical malaria in 1385 study participants aged 10+ years stratified by the percentage of children aged 0–9 years in the same household who received one or more rounds of SMC, or the mean number of SMC rounds per child, during the 2021 malaria transmission season in the Upper River Region of The Gambia

SMC coverage	Incidence rates per 100 person-months (cases/PM ^a)	Unadjusted HR (95% CI) p	Fully adjusted HR (95% CI) p
% Children in household re	eceived at least 1 round SMC		
25%	5.46 (24/439)	1 (ref)	1 (ref)
25–75%	2.51 (90/3585)	0.52 (0.31, 0.88) 0.015	0.60 (0.34, 1.07) 0.084
80%	1.49 (51/3434)	0.29 (0.17, 0.51) < 0.001	0.49 (0.28, 0.86) 0.014
SMC Rounds/child in hous	ehold		
<0.3	5.47 (24/439)	1 (ref)	1 (ref)
0.3–<2	2.63 (88/3344)	0.54 (0.32, 0.91) 0.020	0.57 (0.33, 0.99) 0.047
>2-4	1.44 (53/3675)	0.29 (0.17, 0.51) < 0.001	0.53 (0.28, 1.02) 0.059

All models adjusted for village ID, household size, ratio of children to adults, age in years, household ID, baseline (dry season) infection prevalence in household and % of nights household used insecticide treated nets.

^aIncidence rate per 100 person months of follow-up (total malaria cases/total person months of follow-up).

identified a significant reduction in clinical infections in children 0–9 years, but only decreased asymptomatic infections in children with 3 or more rounds of SMC which did not reach significance (Supplementary File 1 STable 4). Interestingly children who received only 1–2 rounds of SMC benefitted in terms of reduced incidence of clinical episodes but had increased risk of asymptomatic infections compared to children receiving no SMC, though overall prevalences were low (<6.5%). Children who had received only 1–2 SMC round were also more likely to have received SP in the first half of the season (Supplementary File 1 STable 5) perhaps leading to a slightly elevated risk of infection in the following months of rising transmission due to naïve immunity—however, given low prevalence estimates we cannot rule out the potential presence of sampling error. The evidence for the transmission-blocking potential of SP-AQ is conflicting; in vivo and in vitro studies have indicated exposure to SP may increase commitment of blood stage parasites to sexual differentiation resulting in increased gametocyte concentrations and potential increased transmission potential to feeding mosquitoes^{40–42} however detailed analysis of this phenomenon suggests impacts occur for a small number of drug classes under a narrow drug concentration window, and are unlikely to result in a net increase in transmission potential⁴³. Pyrimethamine has also shown suppressive activity against oocyte production or successful development of sporozoites in the mosquito^{44,45}, which may offset increased gametocytogenesis following SP use or SMC but are also characterised by small sample sizes and incomplete accounting for potential confounding, and alone do not resolve this conflict^{46,47}. Whilst gametocyte concentrations were



Fig. 4 | Prevalence and density of asymptomatic infections in 1,056 study participants aged 10+ years surveyed between the 27th Sept and 27th Nov 2021, by household-level coverage of SMC in young children. a Prevalence of asymptomatic infection in all participants aged 10+ years, by the coverage of SMC in young children in the same household. SMC coverage is defined as the percentage of children 0–9 years who received at least one round of SMC in three groups: <25% (blue bar), 25–75% (orange bar) and 80%+ (green bar). b Kernel density plot showing the log (natural) parasite concentrations originally measured in counts per µL in PCR-positive participants aged 10+ years. Plots are stratified by SMC coverage in children 0–9 years in the same household, defined as the percentage of children 0–9 years who received at least one round of SMC in three groups: <25% (blue plot), 25–75% (orange plot) and 80%+ (green plot). Median parasite concentrations for each SMC coverage group are shown by dotted vertical lines (colours to match SMC



Article

coverage category); cut-offs for high-density infections (averaged for age groups 10–15 and 16+ years) are shown as solid black lines. c Prevalence of asymptomatic infection in all participants aged 10+ years, by the coverage of SMC in young children in the same household. SMC coverage is defined as the mean number of SMC rounds per child also in three groups: <0.3 (blue bar), 0.3–2 (orange bar) and >2 (green bar). d Kernel density plot showing the log (natural) parasite concentrations originally measured in counts per μ L in PCR-positive participants aged 10+ years. Plots are stratified by SMC coverage in children 0–9 years in the same household, defined as the mean number of SMC rounds per child also in three groups: <0.3 (blue plot), 0.3–2 (orange plot) and >2 (green plot). Median parasite concentrations for each SMC coverage group are shown by dotted vertical lines (colours to match SMC coverage category); cut-offs for high-density infections (averaged for age groups 10–15 and 16+ years) are shown as solid black lines.

Table 4 | Prevalence of asymptomatic infections in participants 10+ years 27th Sept-29th Nov 2021, by household coverage of SMC in young children

Household SMC coverage	Prevalence of asym	ptomatic infection		Prevalence of high parasite density infections ^a				
	<i>n/N</i> (% prevalence)	Unadjusted RR (95% CI) p	Fully adjusted RR (95% CI) p	n/N (% prevalence)	Unadjusted RR (95% CI) <i>p</i>	Fully adjusted RR (95% CI) p		
% children in household	d received at least 1 ro	und SMC						
25%	12/64 (18.8%)	1 (ref)	1 (ref)	3/12 (25.0%)	1 (ref)	1 (ref)		
25–75%	45/503 (9.0%)	0.34 (0.10–1.15) 0.082	0.44 (0.20–0.96) 0.040	26/45 (57.8%)	2.31 (0.84–6.36) 0.104	4.30 (0.76–25.28) 0.099		
80%	34/489 (7.0%)	0.28 (0.08–0.93) 0.039	0.52 (0.22–1.18) 0.117	22/34 (64.7%)	2.59 (0.94–7.12) 0.065	4.94 (0.80–28.31) 0.086		
Household mean number SMC rounds/child								
<0.3	12/64 (18.8%)	1 (ref)	1 (ref)	3/12 (25.0%)	1 (ref)	1 (ref)		
0.3-<2	44/477 (9.2%)	0.35 (0.10–1.15) 0.084	0.40 (0.19–0.93) 0.032	26/44 (59.1%)	2.36 (0.86–6.49) 0.095	4.77 (0.91–27.46) 0.080		
>2-4	35/515 (6.8%)	0.27 (0.08–0.93) 0.037	0.58 (0.24–1.39) 0.220	22/35 (62.9%)	2.51 (0.91–6.92) 0.074	5.74 (0.92–35.68) 0.061		

Adjusted models included village ID, household size, ratio of children to adults, age in years, household ID, baseline (dry season) infection prevalence in household, week of cross-sectional visit and % of nights during season household used insecticide-treated nets when asked as fixed effects.

^aParticipants with natural log parasite concentrations in the top 50% for their age group (0-4, 5-9, 10-15, 16+ years) were categorised as high-density infections.

not available for participants in our study, we did observe a trend of increased parasite concentrations in children with SMC and in older participants in households with higher SMC coverage known to be positively associated with gametocyte density¹², however, infection rates were too small to confirm this relationship. Other studies have shown that densities of both asexual and gametocyte stages are nonetheless considerably lower in asymptomatic infections compared to clinical disease^{48,49}. This current body of evidence highlights the mechanisms of action of SP-AQ SMC are complex and not yet fully elucidated; the net product of a trade-off between reduced clinical infection, increased asymptomatic parasitaemia and

gametocytaemia and potential suppressive effects on mosquito lifecycle stages may be a reduction in overall transmission potential.

An assessment of the transmission-reducing effects of SMC is most useful for policy if based on results of routinely delivered programmes. This setting better predicts real-life impacts under sub-optimal or heterogenous SMC coverage and in the presence of other nationally implemented malaria interventions. High intervention coverage and strict eligibility criteria for participants and clusters observed in trial settings often do not transfer with high fidelity or effectiveness for this reason. Nonetheless, a limitation of our observational approach is determining the appropriate techniques to account for other exogenous predictors of transmission that may also differ by SMC coverage, not all of which can be measured. We included a priori fixed effects in statistical models to address this. Village ID was used as a proxy for underlying transmission potential for a community, clustering by ethnicity and in the context of the wider INDIE study also allowed us to adjust for ongoing interventions that differed by community. We also included household-level baseline malaria infection prevalence and indicators of individual or household level bed net usage - variation in treated bed net usage can predict mosquito biting rates, local intensity of malaria transmission and/or potential to adhere to other protective behaviours and may correlate with higher acceptability and adherence to SMC programmes. We examined other characteristics of households by SMC coverage group (Supplementary Data 1, 2) to identify potentially confounding characteristics. All else being equal, the household impact of SMC may be influenced by the number of children relative to older inhabitants and/or total household size. These parameters differed by SMC coverage level in this population. We, therefore, addressed these sources of potential confounding by presenting both unadjusted models and models adjusted for the above features with correction of errors to account for clustering at person or household level. We excluded some households from our spatial analysis sample - which detected no geographical clustering of effects of householdlevel models of SMC - however the vast majority of households (90%, 129/ 144) across the site were successfully enumerated and the missing (15 households) were spread amongst all communities (Supplementary Fig. 2). Given P values for Moran tests were close to 1 it is unlikely data from the 15 excluded households would significantly modify these. As a robustness test, we defined coverage of SMC in two ways, as the percentage of children receiving any SMC, and the mean number of SMC rounds per child with both definitions showing similar patterns of impact. There may be some misclassification of SMC status as the NMCP did not have a policy of directly observed treatment for SMC at the time, and the status for a third of children was based only on caregiver recall. However given the short length of the transmission season and the short length of time between our surveys and the SMC rounds themselves, we expect the levels of misclassification to be relatively low. Our study did not allow us to separate differential herd impacts of SMC coverage by age (0-4 years and 5-9 years), highlighting a potential focus for further research. We also did not perform a detailed assessment of additional herd impacts of household SMC coverage on eligible children irrespective of their own status. In both cases malaria outcomes were relatively less common in this group and in the case of the second initially significant positive effects did not survive adjustment for key confounders. Given wide confidence intervals (STable 6) including the potential for both very large reductions or increases in hazard of clinical malaria, it is likely that this sub-analysis is underpowered. Therefore the possible impacts in this age group therefore remain to be established in future studies. Whilst a cost analysis was outside the scope of this study, estimates from the Senegal study where SMC was extended to the same age range suggest the programme can be implemented with an average delivery cost of \$0.50 per child per month⁶, however, additional context-specific data of extended programmes from other sites would be useful for policy. Our study was conducted in a moderate/low transmission setting-future studies in higher burden settings will add to this growing evidence base.

In conclusion, we show in the context of a routine extended-age chemoprevention programme with SP-AQ that SMC is associated with a significant reduction in clinical malaria in direct recipients and with a significant reduction in clinical and asymptomatic infection in older household inhabitants. Our data demonstrate these effects are robust to adjustment, are household-specific with little negligible from nearby households. The results support findings of previous laboratory and simulation studies of the importance of children to the infectious reservoir for malaria, addressing a critical evidence gap in demonstrating herd effects in practice. They point to important additional benefits of SMC in reducing overall malaria transmission, highlighting such programmes are likely to be more cost-effective than currently estimated.

Data availability

Study data (for Tables and in-line results) will be made publicly accessible after the publication of the impact paper for the INDIE 1b project. Datasets will be available via the LSHTM Data Compass data repository (https://datacompass.lshtm.ac.uk/), or from the corresponding authors upon reasonable request. The source data behind the graphs in the figures can be found in Supplementary Data 3.

Received: 8 August 2023; Accepted: 18 April 2024; Published online: 22 May 2024

References

- WHO. WHO Policy Recommendation: Seasonal Malaria Chemoprevention (SMC) for Plasmodium falciparum malaria control in highly seasonal transmission areas of the Sahel sub-region in Africa March 2012. https://www.who.int/malaria/areas/preventive_therapies/ children/en/ (2012).
- Cairns, M. et al. Effectiveness of seasonal malaria chemoprevention (SMC) treatments when SMC is implemented at scale: case-control studies in 5 countries. *PLoS Med.* 18, e1003727 (2021).
- Baba, E. et al. Effectiveness of seasonal malaria chemoprevention at scale in west and central Africa: an observational study. *Lancet* 396, 1829–1840 (2020).
- 4. WHO. WHO Consolidated Guidelines for Malaria (World Health Organization, 2022).
- Okiro, E. A. et al. Age patterns of severe paediatric malaria and their relationship to Plasmodium falciparum transmission intensity. *Malar. J.* 8, 4 (2009).
- Cissé, B. et al. Effectiveness of seasonal malaria chemoprevention in children under ten years of age in Senegal: a stepped-wedge clusterrandomised trial. *PLoS Med.* **13**, e1002175 (2016).
- WHO. Updated WHO recommendations for malaria chemoprevention among children and pregnant women. https://www.who.int/news/ item/03-06-2022-Updated-WHO-recommendations-for-malariachemoprevention-among-children-and-pregnant-women (2022).
- Diawara, S. I. et al. Effect of seasonal malaria chemoprevention in children between 5 and 9 years old in Kita and Bafoulabe districts, Mali. *Parasite Epidemiol. Control* 18, e00258 (2022).
- Konaté, D. et al. Effectiveness and community acceptance of extending seasonal malaria chemoprevention to children 5 to 14 years of age in Dangassa, Mali. *Am. J. Trop. Med. Hyg.* **106**, 648–654 (2022).
- Moukénet, A. et al. Extending delivery of seasonal malaria chemoprevention to children aged 5–10 years in Chad: a mixedmethods study. *Glob. Health Sci. Pract.* **10**, e2100161 (2022).
- Stone, W., Gonçalves, B. P., Bousema, T. & Drakeley, C. Assessing the infectious reservoir of falciparum malaria: past and future. *Trends Parasitol* 31, 287–296 (2015).
- 12. Andolina, C. et al. Sources of persistent malaria transmission in a setting with effective malaria control in eastern Uganda: a longitudinal, observational cohort study. *Lancet Infect. Dis.* **21**, 1568–1578 (2021).
- Githeko, A. K. et al. The reservoir of Plasmodium falciparum malaria in a holoendemic area of western Kenya. *Trans. R. Soc. Trop. Med. Hyg.* 86, 355–358 (1992).

- Gonçalves, B. P. et al. Examining the human infectious reservoir for Plasmodium falciparum malaria in areas of differing transmission intensity. *Nat. Commun.* 8, 1133 (2017).
- Rek, J. et al. Asymptomatic school-aged children are important drivers of malaria transmission in a high endemicity setting in Uganda. *J. Infect. Dis.* 226, 708–713 (2022).
- Coalson, J. E. et al. Simulation models predict that school-age children are responsible for most human-to-mosquito Plasmodium falciparum transmission in southern Malawi. *Malar. J.* 17, 147 (2018).
- Tagbor, H. et al. Seasonal malaria chemoprevention in an area of extended seasonal transmission in Ashanti, Ghana: an individually randomised clinical trial. *Trop. Med. Int. Health* **21**, 224–235 (2016).
- McCann, R. S., Cohee, L. M., Goupeyou-Youmsi, J. & Laufer, M. K. Maximizing impact: can interventions to prevent clinical malaria reduce parasite transmission? *Trends Parasitol.* 36, 906–913 (2020).
- 19. World Health Organization. *Global Technical Strategy for Malaria* 2016–2030 (World Health Organization, 2015).
- Binka, F. N., Indome, F. & Smith, T. Impact of spatial distribution of permethrin-impregnated bed nets on child mortality in rural northern Ghana. *Am. J. Trop. Med. Hyg.* **59**, 80–85 (1998).
- Hawley, W. A. et al. Community-wide effects of permethrin-treated bed nets on child mortality and malaria morbidity in western Kenya. *Am. J. Trop. Med. Hyg.* 68, 121–127 (2003).
- Howard, S. C. et al. Evidence for a mass community effect of insecticide-treated bednets on the incidence of malaria on the Kenyan coast. *Trans. R. Soc. Trop. Med. Hyg.* 94, 357–360 (2000).
- Conteh, L. et al. Costs and cost-effectiveness of malaria control interventions: a systematic literature review. *Value Health* 24, 1213–1222 (2021).
- Winskill, P., Walker, P. G., Cibulskis, R. E. & Ghani, A. C. Prioritizing the scale-up of interventions for malaria control and elimination. *Malar. J.* 18, 122 (2019).
- 25. SMC Alliance. SMC Impact project in the Gambia. SMC, https://www. smc-alliance.org/smc-impact-project-in-the-gambia (2021).
- Stresman, G. H. et al. Do hotspots fuel malaria transmission: a villagescale spatio-temporal analysis of a 2-year cohort study in The Gambia. *BMC Med* 16, 160 (2018).
- 27. Stresman, G. H. et al. Focal screening to identify the subpatent parasite reservoir in an area of low and heterogeneous transmission in the Kenya highlands. *J. Infect. Dis.* **212**, 1768–1777 (2015).
- London School of Hygiene and Tropical Medicine. P. Falciparum infection dynamics and transmission to inform elimination. https:// clinicaltrials.gov/ct2/show/NCT04053907 (2022).
- 29. Mwesigwa, J. et al. On-going malaria transmission in The Gambia despite high coverage of control interventions: a nationwide cross-sectional survey. *Malar. J.* **14**, 314 (2015).
- Kirby, M. J. et al. Risk factors for house-entry by malaria vectors in a rural town and satellite villages in The Gambia. *Malar. J.* 7, 2 (2008).
- Caputo, B. et al. Anopheles gambiae complex along The Gambia River, with particular reference to the molecular forms of An. gambiae s.s. *Malar. J.* 7, 182 (2008).
- ACCESS-SMC Partnership Effectiveness of seasonal malaria chemoprevention at scale in west and central Africa: an observational study. *Lancet Lond. Engl.* **396**, 1829–1840 (2020).
- Boerma, T., AbouZahr, C., Evans, D. & Evans, T. Monitoring intervention coverage in the context of universal health coverage. *PLoS Med.* **11**, e1001728 (2014).
- Requejo, J. et al. Assessing coverage of interventions for reproductive, maternal, newborn, child, and adolescent health and nutrition. *BMJ* 368, 16915 (2020).
- 35. Hayes, R. & Moulton, L. *Cluster Randomised Trials* (Chapman and Hall/CRC, 2021).
- Eldridge, S. & Kelly, S. Trial reporting. in A Practical Guide to Cluster Randomised Trials in Health Services Research 218–265 (John Wiley & Sons, Ltd, 2012). https://doi.org/10.1002/9781119966241.ch10.

- Richardson, S. et al. Modeled impact of seasonal malaria chemoprevention on district-level suspected and confirmed malaria cases in chad based on routine clinical data (2013–2018). *Am. J. Trop. Med. Hyg.* **105**, 1712–1721 (2021).
- Diawara, F. et al. Measuring the impact of seasonal malaria chemoprevention as part of routine malaria control in Kita, Mali. *Malar.* J. 16, 325 (2017).
- 39. Bousema, T. et al. Hitting hotspots: spatial targeting of malaria for control and elimination. *PLoS Med* **9**, e1001165 (2012).
- 40. Peatey, C. L. et al. Effect of antimalarial drugs on plasmodium falciparum gametocytes. *J. Infect. Dis.* **200**, 1518–1521 (2009).
- Portugaliza, H. P., Llorà-Batlle, O., Rosanas-Urgell, A. & Cortés, A. Reporter lines based on the gexp02 promoter enable early quantification of sexual conversion rates in the malaria parasite Plasmodium falciparum. *Sci. Rep.* 9, 14595 (2019).
- Puta, C. & Manyando, C. Enhanced gametocyte production in Fansidar-treated Plasmodium falciparum malaria patients: implications for malaria transmission control programmes. *Trop. Med. Int. Health* 2, 227–229 (1997).
- 43. Thommen, B. T. et al. Revisiting the effect of pharmaceuticals on transmission stage formation in the malaria parasite Plasmodium falciparum. *Front. Cell. Infect. Microbiol.* **12**, 802341 (2022).
- Villa, M., Buysse, M., Berthomieu, A. & Rivero, A. The transmissionblocking effects antimalarial drugs revisited: mosquito fitness costs and sporontocidal effects of artesunate and sulfadoxine-pyrimethamine. *Int. J. Parasitol.* **51**, 279–289, https://doi.org/10.1101/2020.05.19. 103408 (2021).
- White, N. J. Malaria. in *Manson's Tropical Infectious Diseases*, 23rd edn, 532–600.e1 (W.B. Saunders, 2014). https://doi.org/10.1016/ B978-0-7020-5101-2.00044-3.
- Ahmad, A. et al. Gametocyte carriage after seasonal malaria chemoprevention in Plasmodium falciparum infected asymptomatic children. *Malar. J.* 20, 169 (2021).
- 47. Barnes, K. I. et al. Increased gametocytemia after treatment: an early parasitological indicator of emerging sulfadoxine-pyrimethamine resistance in falciparum malaria. *J. Infect. Dis.* **197**, 1605–1613 (2008).
- Bousema, T. & Drakeley, C. Epidemiology and Infectivity of Plasmodium falciparum and Plasmodium vivax gametocytes in relation to malaria control and elimination. *Clin. Microbiol. Rev.* 24, 377–410 (2011).
- Collins, K. A. et al. A cohort study on the duration of Plasmodium falciparum infections during the dry season in The Gambia. *J. Infect. Dis.* 226, 128–137 (2022).

Acknowledgements

We would like to acknowledge the contribution of all the fieldworkers and participants of the 10 Upper River Region communities who provided data for this study; Isabel Byrne for advice on the spatial analysis approach, and Professor Paul Milligan for information on the status of extended-age SMC pilots in West Africa. This study was funded via a grant to C.D. and T.B. from the Bill and Melinda Gates Foundation (grant ref: OPP1173572). The Foundation did not provide any input to any aspect of the design of the study or development of this manuscript.

Author contributions

S.S., B.C., U.D.A., T.B., A.E., M.M. and C.D.—responsible for study concept and design. B.C., M.M., A.E., T.B., C.D.—responsible for fieldwork coordination. A.N., S.S., M.M.—responsible for data management. H.M.S.—responsible for entomological surveillance data collection. B.E., M.O.N.—responsible for laboratory assays. S.S., J.B.—responsible for statistical analysis. S.S. wrote the first draft of the paper and all authors reviewed and approved the manuscript.

Competing interests

The authors declare no competing interests.

Additional information

Supplementary information The online version contains supplementary material available at https://doi.org/10.1038/s43856-024-00503-0.

Correspondence and requests for materials should be addressed to Seyi Soremekun or Chris Drakeley.

Peer review information *Communications Medicine* thanks Pfupajena Machingauta and the other, anonymous, reviewer(s) for their contribution to the peer review of this work. A peer review file is available.

Reprints and permissions information is available at http://www.nature.com/reprints

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

© The Author(s) 2024